Author's response to reviews

Title: Reducing resistance to treatment, through group intervention, improves clinical measurements in patients with type 2 diabetes.

Authors:

Liora Valinsky (liora.v@meuhedet.co.il)
Rachel PREISS (preiss_r@mac.org.il)
Ronit Endevelt (rendevelt@univ.haifa.ac.il)
Keren Dopelt (dopkm@013.net)
Moshe Mishali (moshiko2@netvision.net.il)
Anthony David Heymann (tonyheymann@gmail.com)

Version: 5 Date: 16 December 2013

Author's response to reviews:

December 15, 2013

Dr Elizabeth Beverly
Editor in chief
BMC Endocrine Disorders.

Dear Editor,


We thank you again for reviewing our manuscript. The two editor’s comments were implemented in the manuscript: Table 2 was added; an important sentence was added to the limitations. Please see marked in yellow.

We thank you again and would appreciate your reconsideration.

Sincerely,

Liora Valinsky
Director of Clinical Quality
Meuhedet Health Care Services
Tel Aviv, Israel